Author's response to reviews

Title: Cost-effectiveness of fibronectin testing in a triage in women with threatened preterm labor: alleviation of pregnancy outcome by suspending tocolysis in early labor (APOSTEL-I trial).

Authors:

Jolande Y Vis (j.y.vis@amc.nl)
Femke F Wilms (femke_wilms@hotmail.com)
Martijn A Oudijk (m.a.oudijk@amc.uva.nl)
Martina M Porath (m.porath@mmc.nl)
Liesbeth CJ Scheepers (hcj.scheepers@mumc.nl)
Kitty WM Bloemenkamp (k.w.m.bloemenkamp@lumc.nl)
Annemiek C Bolte (ac.bolte@vumc.nl)
Jérôme Cornette (j.cornette@erasmusmc.nl)
Jan B Derks (j.b.derks@umcutrecht.nl)
Johannes J Duvekot (j.j.duvekot@erasmusmc.nl)
Jim van Eyck (j.van.eyck@isala.nl)
Anneke Kwee (a.kwee@umcutrecht.nl)
Brent C Opmeer (b.c.opmeer@amc.uva.nl)
Marielle van Pampus (m.van.pampus@og.umcg.nl)
Fred K Lotgering (f.lotgering@obgyn.umcn.nl)
Sicco A Scherjon (s.scherjon@lumc.nl)
Krystyna M Sollie (k.m.sollie@og.umcg.nl)
Marc EA Spaanderman (m.spaanderman@obgyn.umcn.nl)
Christine Willekes (c.willekes@mumc.nl)
Joris AM van der Post (j.a.vanderpost@amc.uva.nl)
Ben Willem J Mol (b.w.mol@amc.uva.nl)

Version: 4  Date: 25 August 2009

Author's response to reviews:

Dear Editor,

Please find the latest version of the APOSTEL-1 protocol. We have proofread the article, corrected some typographical and grammatical errors and included a legend for figure 1 on your request.

Sorry for our delay in the payment of the processing charge, I will make sure you receive the payment as soon as possible.

Thank you for publishing this article, I your have any further comments, please let me know.

Yours sincerely,
Jolande Vis